

# Modeling the impact of changes in day-care contact patterns on the dynamics of varicella transmission in France between 1991 and 2015

Valentina Marziano, Piero Poletti, Guillaume Béraud, Pierre-Yves Boëlle, Stefano Merler, Vittoria Colizza

### ▶ To cite this version:

Valentina Marziano, Piero Poletti, Guillaume Béraud, Pierre-Yves Boëlle, Stefano Merler, et al.. Modeling the impact of changes in day-care contact patterns on the dynamics of varicella transmission in France between 1991 and 2015. PLoS Computational Biology, 2018, 14 (8), pp.e1006334. 10.1371/journal.pcbi.1006334. hal-01880120

# HAL Id: hal-01880120 https://hal.sorbonne-universite.fr/hal-01880120v1

Submitted on 24 Sep 2018  $\,$ 

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License

RESEARCH ARTICLE

Modeling the impact of changes in day-care contact patterns on the dynamics of varicella transmission in France between 1991 and 2015

Valentina Marziano<sup>1</sup>\*, Piero Poletti<sup>1</sup>, Guillaume Béraud<sup>2,3,4</sup>, Pierre-Yves Boëlle<sup>5</sup>, Stefano Merler<sup>1‡</sup>, Vittoria Colizza<sup>5‡</sup>

1 Center for Information Technology, Bruno Kessler Foundation, Trento, Italy, 2 Médecine Interne et Maladies Infectieuses, Centre Hospitalier de Poitiers, Poitiers, France, 3 EA2694, Université Droit et Santé Lille 2, Lille, France, 4 Interuniversity Institute for Biostatistics and statistical Bioinformatics, Hasselt University, Hasselt, Belgium, 5 INSERM, Sorbonne Université, Institut Pierre Louis d'Epidémiologie et de Santé Publique IPLESP, Paris, France

‡ These authors are joint senior authors on this work. \* marziano@fbk.eu

## Abstract

Annual incidence rates of varicella infection in the general population in France have been rather stable since 1991 when clinical surveillance started. Rates however show a statistically significant increase over time in children aged 0-3 years, and a decline in older individuals. A significant increase in day-care enrolment and structures' capacity in France was also observed in the last decade. In this work we investigate the potential interplay between an increase of contacts of young children possibly caused by earlier socialization in the community and varicella transmission dynamics. To this aim, we develop an age-structured mathematical model, informed with historical demographic data and contact matrix estimates in the country, accounting for longitudinal linear increase of early childhood contacts. While the reported overall varicella incidence is well reproduced independently of mixing variations, age-specific empirical trends are better captured by accounting for an increase in contacts among pre-school children in the last decades. We found that the varicella data are consistent with a 30% increase in the number of contacts at day-care facilities, which would imply a 50% growth in the contribution of 0-3y old children to overall yearly infections in 1991–2015. Our findings suggest that an earlier exposure to pathogens due to changes in day-care contact patterns, represents a plausible explanation for the epidemiological patterns observed in France. Obtained results suggest that considering temporal changes in social factors in addition to demographic ones is critical to correctly interpret varicella transmission dynamics.

### Author summary

During the last decades, an increasing circulation of varicella in the early childhood has been observed in France. A plausible explanation of this trend may rely on the progressive



# 

**Citation:** Marziano V, Poletti P, Béraud G, Boëlle P-Y, Merler S, Colizza V (2018) Modeling the impact of changes in day-care contact patterns on the dynamics of varicella transmission in France between 1991 and 2015. PLoS Comput Biol 14(8): e1006334. https://doi.org/10.1371/journal. pcbi.1006334

**Editor:** Matthew (Matt) Ferrari, The Pennsylvania State University, UNITED STATES

Received: December 18, 2017

Accepted: June 27, 2018

Published: August 1, 2018

**Copyright:** © 2018 Marziano et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All data and code used in this study are available from the Figshare digital repository at: https://figshare.com/s/ 4e7f05ea4154c1db8dbc (DOI: 10.6084/m9. figshare.6344876).

Funding: VM, PP and SM received funding from the European Commission Horizon2020 CIMPLEX project, grant agreement N.641191 (https://ec. europa.eu/programmes/horizon2020/). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** I have read the journal's policy and the authors of this manuscript have no competing interests in relation to this work. Outside of this work, PP declares personal fees from Merck & Co. Inc. for consultancy activity.

increase of day-care attendance in the past years, which could have anticipated the exposure of young children to the infection. We propose a retrospective modelling study to assess whether the varicella dynamics in France since 1991 can be explained in terms of increasing day-care contacts of children under 3 years of age. To this aim, we develop a model including demographic changes and variations in age-specific contact rates over time. Our findings suggest that a 30% increase of day-care contacts in early childhood can explain the observed epidemiological trends. Obtained results highlight that temporal changes in contact patterns can significantly affect the transmission of childhood infectious diseases and should therefore be considered when investigating medium and longterm epidemiological patterns. A better understanding of the interplay between changing social behavior and disease transmission can help the interpretation of surveillance data and the design of effective and targeted intervention strategies.

### Introduction

Varicella is a vaccine-preventable infectious disease caused by exposure to Varicella-Zoster Virus (VZV). The pathogen is antigenically stable so that, in principle, no changes in transmission or immunogenicity caused by mutations of the virus are expected over time [1]. In France, about 90% of the population gets infected with varicella before 8 years of age; most of infections occur in the early childhood and result in relatively mild symptoms [2,3]. In this country, vaccination against varicella is not recommended and little used in children. Previous studies have shown that temporal changes in the crude birth rate of a population are key drivers of the dynamics of childhood infectious diseases, such as varicella and measles, by affecting the replenishment of susceptible individuals in the population [4-8]. Since the early 90s, France experienced a roughly constant crude birth rate [9] after a strong demographic transition in the last century characterized by a progressive decline of birth and death rates. This corresponded to stable varicella infection rates at the population level between 1991 and 2015, as revealed by the French GPs Sentinelles Network for surveillance [10]. However, when looking at the distribution of cases by age, surveillance data highlight that during this period varicella incidence has increased in children aged 0-3 years and decreased in children aged 4-7 years. A similar pattern has been detected in other countries, including Slovenia, the US and England [11–16], where varicella incidence doubled in children aged 0–4 years between 1983 and 1998 and halved in those aged 5-14 years. This suggests that, beyond changes in fertility and mortality rates, other factors may influence the circulation of childhood infections across the different age segments of the population. One of them is variations in the population mixing patterns driven by socio-demographic changes, affecting school attendance and household structure. In particular, some epidemiological studies have hypothesized that an increase of varicella incidence in young ages may be ascribable to increased social contacts in these age groups, possibly caused by earlier inclusion in nurseries or day-care centers [11–13]. Past modeling efforts, based on a theoretical framework assuming a stationary age distribution of the population, have suggested that a substantial increase of contact rates in preschool children is consistent with the increase in varicella consultations in UK observed between 1970 and 1998 in this age segment [17].

The aim of this work is to assess whether changes in the age-specific varicella incidence observed in France can be the result of an increase of contacts in the early childhood.

To this aim, we considered two transmission models with the same demographic and epidemiological structure, but differing in mixing patterns over time. Both models take explicitly into account demographic changes occurred during the last century [9]. In the first model, mixing patterns between individuals of different ages are assumed to remain constant over time and are modelled according to the age-specific contact matrix estimated for France in 2012 [18]. In the second model, we assume a linear increase of contact rates occurring at daycare facilities for children under 3 years of age, starting in the decades before 2012.

#### Methods

#### Demographic and epidemiological model

The adopted modelling approach is based on a deterministic age-structured model similar to the one developed by previously published studies to investigate historical dynamics of measles across different countries and varicella in Spain [5–7]. The population, grouped into 1-year age classes (0–89+), is initialized in 1850 at the demographic and epidemiological equilibrium. The latter was obtained by running the transmission model with constant crude birth and mortality rates fixed to those observed in 1850, and by initializing the system with 10 infected individuals in a fully susceptible population. Simulations of varicella dynamics from 1850 to 2015 are obtained by running the model as informed by the yearly variations of birth and age-specific mortality rates provided by the National Institute of Statistics and Economical Studies (INSEE).

The demographic model is validated against the age distribution of the population observed in France during the simulated period (1850–2015) [9]. Realistic mixing patterns by age are modelled using contact matrices estimated for France in 2012 [18]. Age-specific contact matrices are here defined as the average number of unique physical and conversational contacts with individuals of different ages occurring daily, regardless their duration and frequency [18].

The transmission of varicella follows an MSIR model. Briefly, maternal antibodies protect new-borns against varicella infection (M) for 2 months on average [19], after which they become susceptible to varicella infection (S). Susceptible individuals are exposed to a time-and age-dependent force of infection  $\lambda_i(t)$  as follows:

$$\lambda_i(t) = \beta \sum_{j=0}^n C_{ij}(t) \frac{I_j(t)}{N_i(t)}$$
(1)

where *t* and *i* denote time and the individuals' age, respectively; n = 89 years is the maximum age considered in the model,  $I_j(t)/N_j(t)$  is the fraction of individuals of age *j* who are infected at time *t* and  $C_{ij}(t)$  is the contact matrix at time *t*, which is defined as the average number of contacts of an individual of age *i* with individuals of age *j*; finally, under the social contact hypothesis [20],  $\beta$  represents an age-independent constant proportionality factor driving the contribution of individuals' contacts to the transmission of the infection.

Once recovered, varicella infected individuals (I) acquire life-long immunity against varicella. The generation time of varicella is assumed equal to 3 weeks on average [21].

Epidemiological and demographic transitions occurring within a given year are described by the following set of ordinary differential equations:

$$\begin{split} \dot{M}_{i}(t) &= \delta_{i0}b(t)N(t) - \omega M_{i}(t) - \mu_{i}(t)M_{i}(t) \\ \dot{S}_{i}(t) &= \omega M_{i}(t) - \lambda_{i}(t)S_{i}(t) - \mu_{i}(t)S_{i}(t) \\ \dot{I}_{i}(t) &= \lambda_{i}(t)S_{i}(t) - \gamma I_{i}(t) - \mu_{i}(t)I_{i}(t) \\ \dot{R}_{i}(t) &= \gamma I_{i}(t) - \mu_{i}(t)R_{i}(t) \end{split}$$

$$(2)$$

where b(t) is the yearly crude birth rate,  $\delta_{ij}$  is the Dirac delta function, N(t) is the population size,  $\omega$  is the waning rate of maternal antibodies,  $\mu_i(t)$  is the yearly age-specific mortality rate and  $\gamma$  is the recovery rate from varicella infection. At the end of each year, the age of individuals is incremented by 1.

#### Contact matrices over time

In this work, we consider two variations of the model described in the previous section, which differ in the assumption made to model mixing patterns during the past.

In model M1, at each time *t* the average number of contacts of an individual of age *i* with individuals of age *j* is computed as:

$$C_{ij}(t) = \bar{C}^{S}_{ij} + \bar{C}^{O}_{ij}$$
 (3)

Where  $\bar{C}_{ij}^{s}$  and  $\bar{C}_{ij}^{o}$  are respectively the matrices of contacts within and outside schools estimated for France in 2012 [18]. School contacts of individuals younger than 3 years of age correspond to social interactions occurring at pre-school facilities, such as day-care centers.

In this sense, model M1 assumes no changes in mixing patterns of individuals over time.

In Model M2, we account for potential temporal changes in contact patterns by adding a time-dependent scaling factor to the matrix of school contacts. Specifically, we assume that  $C_{ij}(t) = f_{ij}(t)\bar{C}_{ij}^s + \bar{C}_{ij}^o$  where

$$f_{ij}(t) := \begin{cases} 1 - \alpha(2012 - t) & \text{if } i < 4 \text{ or } j < 4 \\ 1 & \text{elsewhere} \end{cases}$$
(4)

This simple assumption accounts for linear temporal changes of contact rates in children below 3 years of age and aims at illustrating the potential impact of an increased attendance at day-care centers.

By keeping track of the contribution of different age segments and different settings to the age specific force of infection, we disentangle the proportion of varicella infections caused by school contacts (at different school levels, including day care) and compute the infection matrices representing the proportion of varicella cases generated by contacts of susceptible individuals of age *i* with infected individuals of age *j*.

#### Model estimates

Free parameters of the two models were estimated separately through a Markov chain Monte Carlo (MCMC) approach applied to the negative binomial likelihood of the yearly age-specific incidence of varicella observed in France over the period 1991–2015 [10]. The two models have the following free parameters in common: the transmission scale factor ( $\beta$ ), the varicella reporting rate, which is assumed constant over age and time, and the over-dispersion of the negative binomial distribution. Model M2 has an additional free parameter shaping the changes in school contacts of young children ( $\alpha$ ). Models' performances were compared using the Deviance Information Criterion (DIC) and the Akaike Information Criterion (AIC). Further details on model formulation and estimation are provided in S1 Text.

#### Results

#### Data analysis

The analysis of observed varicella incidence by age group reveals a statistically significant increase of infection rates in children less than 3 years old and a significant decrease in those

older than 5 years (see <u>S1 Text</u>). Temporal changes in the infection rates in new-borns, in children of 4 years of age and in total yearly incidence were found to be not statistically significant.

#### Model estimates

According to both measures used to assess model performances, model M2 (DIC: 6508.8; AIC: 6513.5) is able to better represent the data than model M1 (DIC: 6513.4; AIC: 6523.8).

Both models considered in our analysis are able to reproduce changes in the overall size and age distribution of the French population as observed during the last century (details shown in S1 Text). Briefly, in agreement with available demographic records, the simulated population dynamics shows that the progressive decrease in the crude birth rate experienced between 1900 and 2015 led to a significant reduction of the fraction of children in the population. Consistently with previous works [5,6] and in agreement with observations [10], estimates from both models suggest that the total incidence of varicella did not significantly change between 1991 and 2015, when the yearly crude birth rate remained approximately stable (model M2, Fig 1A; model M1 shown in S1 Text). The estimated varicella reporting rate ranges between 90.7% (95% CI: 85.8%-96.0%) in model M1 and 88.6% (95% CI: 84.1%-93.1%) in model M2. The simulated varicella dynamics were validated against an independent dataset, represented by the age-specific VZV serological profile observed in France in 2003 [2]. According to our simulations and consistently with data, in 2003 about 30% of 2-year-old children were immune to varicella and this fraction increases up to 87% by age 7 (model M2, Fig 1B; model M1, in S1 Text). These results suggest that the overall varicella circulation observed in France during the last decades can be explained in terms of the relatively stable crude birth



**Fig 1. Epidemiological validation of Model M2. a** Mean total incidence of varicella (per 1,000 individuals) as estimated by model M2 over the period 1900–2015. Shaded areas represent 95% CI of model estimates. The inset compares the total incidence of varicella predicted by the model in 1991–2015 (blue) to the one observed by the French GPs Sentinelles Network over the same period (red) [10]. Shaded areas represent 95% CI of model estimates (light blue) and of the data (orange). **b** Age-specific VZV seroprevalence as observed in data from France in 2003 [2] (red) and as estimated by model M2 (blue). Shaded areas represent 95% CI of the data as computed by exact binomial test in [2] (orange) and 95% CI of model estimates (light blue).

https://doi.org/10.1371/journal.pcbi.1006334.g001





Fig 2. Incidence of varicella by age (1991–2015). Incidence of varicella by age as observed in France between 1991 and 2015 (red) and as estimated by model M2 (blue). Shaded areas show the 95% CI of data and of model estimates.

https://doi.org/10.1371/journal.pcbi.1006334.g002

and mortality rates characterizing this period and does not depend on possible changes in mixing patterns.

However, trends in the age-specific varicella incidence estimated with the two models are remarkably different. A detailed analysis on the ability of models M1 and M2 in reproducing the observed dynamics is reported in S1 Text. In particular, model M1 that does not account for changes in mixing patterns over time yields stable infection rates in all age groups between 1991 and 2015 (see S1 Text). This model formulation thus fails to capture the changes in the age distribution of varicella cases observed in the period under study. In contrast, model M2, by explicitly taking into account possible changes in the rate of contacts of young children (0–3 years) established in day-care structures in the years prior to 2012, reproduces the observed trends in the age-specific incidence of varicella (see Fig 2). In particular, model M2 estimates a 12.1% increase of varicella incidence in 0–3 years old children over 1991–2015 and a 13% decrease of varicella incidence in older age groups over the same period. The estimated increase in varicella incidence among 0–3 years is the result of a 18% increase in 1-year old



**Fig 3. Infection matrices over time. a** Fraction of varicella cases (%) generated by contacts of susceptible individuals of age i (x axis) with infected individuals of age j (y axis), as estimated by model M2 in the year 1992 (average over 100 simulation runs with the mean estimates of model parameters). **b** The same as **a** but for 2002. **c** The same as **a** but for 2012.

https://doi.org/10.1371/journal.pcbi.1006334.g003

children, 11% in 2-year old children, and only 2% in 3-year old children (Fig 2, first row). The estimated changes of varicella transmission dynamics are ascribable to an increase in the average number of day-care contacts during the last decades. According to our estimates, in 1991, day-care contacts represented on average the 15.5% of the total contacts of children aged 0–3 years, while this fraction increased up to 19.2% in 2012 (details are provided in <u>S1 Text</u>).

An increase in the proportion of infection transmission due to contacts among children aged 0–3 years from 1992 to 2012 is also detectable (Fig 3). Specifically, the estimated contribution of contacts among children 0–3 years to the infection transmission increased between 1992 and 2012 from 19.4% to 28.6%, while that of contacts among children 4–6 years decreased from 24.4% to 20.1%.

The increase of day-care contacts in the early childhood estimated by model M2 has an impact on the relative contribution of different settings to the overall transmission of varicella. According to our results, although the total incidence of varicella slightly decreased between 1991 and 2015 (from 12.1 to 11 cases per 1,000 individuals), the fraction of cases generated at school facilities of any level (i.e. day care, pre-primary and primary schools) raised from about 43.1% to 46.11% over the same period (see Fig 4A). Such increase is mainly driven by the changing role of day-care centers in varicella transmission, whose contribution to the total varicella cases rises from, on average, 9.1% in 1991 to 17.6% in 2015 (Fig 4A). In particular, infections among children under 3 years of age caused by day-care contacts raised from 27.8% in 1991 to 39.1% in 2015 (Fig 4B and S1 Text).

On the other hand, the percentage of the total yearly infections generated in pre-primary schools decreased from 27.8% to 22.8% and the fraction of infections in other settings diminished from 56.9% to 53.9% (Fig 4A). However, the contribution of different settings to infections occurring among individuals older than 4 years remained substantially unchanged (see Fig 4C and 4D).

#### Discussion

Epidemiological data collected in France between 1991 and 2015 show that varicella incidence rates have remained approximately constant following a stabilization of the birth rate in the early 90's [9,10]. This is coherent with our understanding of the role played by dynamics of the crude birth rate in shaping the transmission of childhood infections, such as measles and





https://doi.org/10.1371/journal.pcbi.1006334.g004

varicella [4–7,22]. Data disaggregated by age group however reveal that infection rates have increased in children younger than 3 years, therefore suggesting an increased circulation of varicella in early childhood. Similar patterns have been detected in other regions and countries, e.g. England, Slovenia and the US [11–16]. A plausible explanation for the observed epidemiological trends may rely on the increase of contacts among pre-school children, possibly caused by growing attendance rates at nurseries and day-care centers [11,12,23–25]. In this work, we investigated this hypothesis and showed that a progressive increase of mixing among 0–3 year-old children may have led to a 12% increase of varicella incidence in this age group between 1991 and 2015. Specifically, our results suggest that a 30% growth in the average number of day-care contacts may have increased by 50% the contribution of 0–3 year-old children to the overall number of yearly infections during the considered period. According to model estimates, although in the last decades the fraction of infections generated in schools of any level (day care included) remained rather stable around 45%, the fraction of transmission in day-care centers almost doubled during the same period.

From a policy-making perspective, our results suggest that the inclusion of children in daycare facilities or nurseries is expected to produce an earlier exposure to pathogens, increasing their risk of contracting infectious diseases. These findings may be useful to interpret results of policies (e.g. vaccination) and to design effective and targeted intervention strategies, e.g. scheduling age at vaccination. It is also important to note that a decrease in the age at infection for childhood diseases may be partially masked by rather stable incidence rates in the overall population.

The central role of age at entry in the community in the early childhood phase suggested by our analysis is widely supported by previous works. Silhol et al. [26] showed that the median age at varicella infection may be related to the fraction of children attending pre-schools, thus explaining the large variability in the age-specific seroprevalence observed across European countries [27]. Early exposure in day-care facilities was also suggested as a possible driver for the increase in varicella incidence reported in children from 12 months to 2 years in Slovenia in the period 1979–89 [12]. More generally, several studies identified a clear link between changes in mixing patterns due to the school calendar (e.g. school term vs. school holidays) and the strong seasonality of varicella dynamics [21,28–30]. A modeling study, based on theoretical age-specific contact patterns, suggested that the increase in GP varicella consultations rates observed in the UK between 1970 and 1998 among the youngest age segments of the population was compatible with an increase in early childhood contact rates [17].

Population surveys and modelling approaches have been proposed to investigate human mixing patterns by age, providing static estimates of country-specific contact matrices (e.g. [18,31–35]). The characterization of contacts was indeed found to be particularly important to achieve accurate and reliable modelling results and reduce uncertainties on recommendations for vaccination against varicella [36]. However, little is known to what extent mixing patterns by age may change over time, for instance, as a consequence of socio-demographic and legislative changes.

Our analysis represents a first step in this direction based on recent realistic estimates of age-specific contact rates in France and a rather simple assumption on how contacts in preschool children may have changed as a consequence of increased day-care enrolment rates [24]. The model innovates on previous approaches also by explicitly taking into account the potential impact of demographic changes in shaping temporal changes in varicella circulation in the country [5–7].

The hypothesis of an increase in the number of contacts established by 0-3 years old infants at day-care facilities is supported by the increase in both the enrolment rates and the potential capacity of day-care services observed for this age group in France during the last decades [24,25].

Nonetheless, our study presents some limitations that call for a deeper understanding of temporal changes in mixing patterns to improve our interpretation of medium to long-term trends in the epidemiology of infectious diseases.

First, in our model we assume a linear increase in contacts rates of young children shaped by a unique scaling factor for all age groups. This assumption could be too simplistic to accurately reproduce the considered phenomenon. The inclusion of more flexible functional forms to describe temporal changes in contact rates would possibly improve model accuracy in reproducing varicella incidence over time for some age groups. Future modeling efforts in this direction would certainly benefit from cross-sectional and longitudinal studies showing how social mixing has changed over time.

The underestimation of varicella incidence in 3-year-old children with respect to reported data suggests that our model underestimated either the number of contacts or the transmission events in this age class. This may be partially due to mixed enrolment of children in different

structures, as this age indicates the transition from day-care centers, potentially available for up to 54% of children (data for 2014 [25]), to pre-primary schools (ecoles maternelles), characterized by 100% attendance since 2000 [24]. Also, under the social contact hypothesis [20, 37] that is widely adopted for modeling childhood infections [38–40], here we consider age-specific transmission rates to be proportional to average daily contact rates through a single constant proportionality factor. However, it is possible that considering age-specific proportionality factors and taking into account the duration of social interactions may better describe the VZV transmission across different ages [35,41,42].

In our model, we did not consider the impact of changes of pre-school and school attendance on the school size. A study in the region of Corsica found that the age of varicella infection decreased as school size increased, likely due to an increased number of contacts per individual [43]. Including this aspect into our model is rather challenging because of the very diversified offer for early childhood services in France (different types of day-care centers, qualified nannies, qualified nannies at home, etc.) and the lack of a centralized management and registration [25].

Finally, additional mechanisms such as temporal changes in individual mixing outside schools, e.g. due to changes in household size and composition [44], may also play a role in the transmission dynamics of varicella. For example, a doubling varicella incidence was reported in Slovenia in the period 1979–1998 in 0 years old infants notwithstanding children less than 10 months are not accepted at day-care facilities [12]. Earlier varicella infections due to infected older siblings may explain the observed trends [43].

Our modelling results suggest that changes in mixing patterns at day-care structures represent one plausible component leading to the increase of incidence estimates over time in all corresponding age classes. The performed analysis focused on the epidemiology of varicella in France over the period 1991–2015. However, conclusions of this work may apply to other infections, such as measles and pertussis, and to countries that have undergone an increase in the school enrolment of young children, as is the case of England and Slovenia [11,12].

Future estimates on how the age-specific VZV immunity profiles have changed over time, based for instance on serological surveys conducted in different years, would help to exclude or quantify the contribution of other competing hypotheses to changes in day-care contact patterns in shaping the observed temporal variations in VZV incidence rates. For instance, while a constant reporting rate over time and across different ages was here assumed, changes in reporting behavior of individuals of different ages might have also occurred between 1991 and 2015.

Previous work, however, showed that extrapolated GP surveillance data were estimated to capture more than 96% of varicella cases in the 90's in France [45], suggesting that no considerable improvement in consultation rate is thus possible.

#### Supporting information

**S1 Text. Supporting text.** Supporting text containing methodological details and additional results.

(PDF)

#### Acknowledgments

We thank the French GPs Sentinelles Network for useful discussions.

#### **Author Contributions**

Conceptualization: Valentina Marziano, Piero Poletti, Stefano Merler, Vittoria Colizza.

Data curation: Guillaume Béraud, Pierre-Yves Boëlle.

Formal analysis: Valentina Marziano.

Funding acquisition: Stefano Merler.

Investigation: Valentina Marziano, Guillaume Béraud, Pierre-Yves Boëlle.

Methodology: Valentina Marziano, Piero Poletti, Stefano Merler, Vittoria Colizza.

Project administration: Stefano Merler, Vittoria Colizza.

Resources: Stefano Merler, Vittoria Colizza.

Software: Valentina Marziano.

Supervision: Piero Poletti, Stefano Merler, Vittoria Colizza.

Visualization: Valentina Marziano.

Writing - original draft: Valentina Marziano.

Writing – review & editing: Valentina Marziano, Piero Poletti, Guillaume Béraud, Pierre-Yves Boëlle, Stefano Merler, Vittoria Colizza.

#### References

- Loparev VN, Gonzalez A, Deleon-Carnes M, Tipples G, Fickenscher H, Torfason EG, Schmid DS. Global identification of three major genotypes of varicella-zoster virus: longitudinal clustering and strategies for genotyping. *J Virol.* 2004; 78: 8349–8358. (https://doi.org/10.1128/JVI.78.15.8349-8358.2004) PMID: 15254207
- Khoshnood B, Debruyne M, Lançon F, Emery C, Fagnani F, Durand I, Floret D. Seroprevalence of varicella in the French population. *Pediatr Infect Dis J.* 2006; 25: 41–4. PMID: 16395101
- 3. Preblud SR. Varicella: complications and costs. Pediatrics. 1986; 78: 728–35. PMID: 3093966
- Earn DJ, Rohani P, Bolker BM, Grenfell BT. A simple model for complex dynamical transitions in epidemics. Science. 2000; 287: 667–70. PMID: 10650003
- Merler S, Ajelli M. Deciphering the relative weights of demographic transition and vaccination in the decrease of measles incidence in Italy. *Proc R Soc B*. 2014; 281: 20132676. (https://doi.org/10.1098/ rspb.2013.2676) PMID: 24403333
- Marziano V, Poletti P, Guzzetta G, Ajelli M, Manfredi P, Merler S. The impact of demographic changes on the epidemiology of herpes zoster: Spain as a case study. *Proc R Soc B*. 2015; 282: 20142509. (https://doi.org/10.1098/rspb.2014.2509) PMID: 25761709
- Trentini F, Poletti P, Merler S, Melegaro A. Measles immunity gaps and the progress towards elimination: a multi-country modeling analysis. *Lancet Infect Dis.* 2017; 17: 1089–1097. (https://doi.org/10. 1016/S1473-3099(17)30421-8) PMID: 28807627
- Li S, Ma C, Hao L, Su Q, An Zh, Ma F, Xie S, Xu A, Zhang Y, Ding Z, Li H, Cairns L, Wang H, Luo H, Wang N, Li L, Ferrari MJ. Demographic transition and the dynamics of measles in six provinces in China: A modeling study. *PLOS Med.* 2017; 14: 1–18. (https://doi.org/10.1371/journal.pmed.1002255) PMID: 28376084
- University of California, Berkeley (USA) and Max Planck Institute for Demographic Research (Germany). 2014 Human mortality database. Available from: <a href="http://www.mortality.org">http://www.mortality.org</a>. Accessed: 26 October 2017.
- Réseau Sentinelles, INSERM, UPMC. Available from: <u>http://www.sentiweb.fr</u>. Accessed: 26 October 2017.
- 11. Ross AM, Fleming DM. Chickenpox increasingly affects preschool children. *Commun Dis Public Health.* 2000; 3: 213–5. PMID: <u>11014039</u>
- Socan M, Kraigher A, Pahor. Epidemiology of varicella in Slovenia over a twenty-year period (1979– 1998). Epidemiol Infect. 2001; 126: 279–283. (https://doi.org/10.1017/s0950268801005350) PMID: 11349978
- Fairley CK, Miller E. Varicella-Zoster Virus Epidemiology-A Changing Scene?. J Infect Dis. 1996; 174: S314–9. PMID: 8896538

- 14. Joseph CA, Noah ND. Epidemiology of chickenpox in England and Wales, 1967–85. *Br Med J (Clin Res Ed)*. 1988; 296: 673–6.
- 15. Bramley JC, Jones IG. Epidemiology of chickenpox in Scotland: 1981 to 1998. *Commun Dis Public Health.* 2000; 3: 282–7. PMID: 11280260
- Yawn BP, Yawn RA, Lydick E. Community impact of childhood varicella infections. J Pediatr. 1997; 130: 759–65. PMID: 9152286
- Whitaker HJ, Farrington CP. Infections with varying contact rates: application to varicella. *Biometrics*. 2004; 60:615–23. (https://doi.org/10.1111/j.0006-341X.2004.00210.x) PMID: 15339283
- Béraud G, Kazmercziak S, Beutels P, Levy-Bruhl D, Lenne X, Mielcarek N, Yazdanpanah Y, Boëlle PY, Hens N, Dervaux B. The French connection: the first large population-based contact survey in France relevant for the spread of infectious diseases. *PloS ONE*. 2015; 10: e0133203. (https://doi.org/10.1371/ journal.pone.0133203) PMID: 26176549
- Pinquier D, Gagneur A, Balu L, Brissaud O, Gras Le Guen C, Hau-Rainsard I, Mory O, Picherot G, De Pontual L, Stephan JL, Maple P, Breuer J, Aubert M, Caulin E, Sana C, Pradat P, Soubeyrand B, Reinert P. Prevalence of anti-varicella-zoster virus antibodies in French infants under 15 months of age. *Clin Vaccine Immunol.* 2009; 16: 484–7. (https://doi.org/10.1128/CVI.00397-08) PMID: 19176690
- Wallinga J, Teunis P, Kretzschmar M. Using data on social contacts to estimate age-specific transmission parameters for respiratory-spread of infectious agents. *Am J Epidemiol.* 2006; 164:936–44. (https://doi.org/10.1093/aje/kwj317) PMID: 16968863
- London WP, Yorke JA. Recurrent outbreaks of measles, chickenpox and mumps: I. Seasonal variation in contact rates. Am J Epidemiol. 1973; 98: 453–68. PMID: 4767622
- Geard N, Glass K, McCaw JM, McBryde ES, Korb KB, Keeling MJ, McVernon J. The effects of demographic change on disease transmission and vaccine impact in a household structured population. *Epidemics*. 2015; 13: 56–64. (https://doi.org/10.1016/j.epidem.2015.08.002) PMID: 26616042
- Organisation for Economic Co-operation and Development (OECD) and Directorate of Employment, Labour and Social Affairs. OECD Family Database. Enrolment in childcare and pre-school. Available from: <u>http://www.oecd.org/els/soc/PF3\_2\_Enrolment\_childcare\_preschool.pdf</u>. Accessed: 26 October 2017.
- 24. Ministère de l'Éducation nationale, de l'Enseignement supérieur et de la Recherche. Infant school in France. Available from: http://cache.media.eduscol.education.fr/file/Europe\_et\_international/21/3/ ecole-maternelle\_113213.pdf. Accessed: 26 October 2017.
- 25. Observatoire national de la petite enfance. Rapport 2016. L'accueil du jeune enfant en 2015. Available from: http://www.caf.fr/sites/default/files/cnaf/Documents/Dser/observatoire\_petite\_enfance/aje\_2016bd.pdf. Accessed: 26 October 2017.
- Silhol R, Boëlle PY. Modelling the effects of population structure on childhood disease: the case of varicella. *PLoS Comput Biol.* 2011; 7: e1002105. (<u>https://doi.org/10.1371/journal.pcbi.1002105</u>) PMID: 21814504
- Nardone A, De Ory F, Carton M, Cohen D, Van Damme P, Davidkin I, Rota MC, De Melker H, Mossong J, Slacikova M, Tischer A. The comparative sero-epidemiology of varicella zoster virus in 11 countries in the European region. *Vaccine*. 2007; 25: 7866–72. (https://doi.org/10.1016/j.vaccine.2007.07.036) PMID: 17919788
- Jackson C, Mangtani P, Fine P, Vynnycky E. The effects of school holidays on transmission of varicella zoster virus, England and Wales, 1967–2008. *PloS ONE*. 2014; 9: e99762. (https://doi.org/10.1371/ journal.pone.0099762) PMID: 24932994
- Metcalf CJ, Bjørnstad ON, Grenfell BT, Andreasen V. Seasonality and comparative dynamics of six childhood infections in pre-vaccination Copenhagen. *Proc R Soc B.* 2009; 276: 4111–8. (https://doi.org/ 10.1098/rspb.2009.1058) PMID: 19740885
- Deguen S, Thomas G, Chau NP. Estimation of the contact rate in a seasonal SEIR model: application to chickenpox incidence in France. *Stat Med.* 2000; 19: 1207–16. PMID: 10797517
- Mossong J, Hens N, Jit M, Beutels P, Auranen K, Mikolajczyk R, Massari M, Salmaso S, Tomba GS, Wallinga J, Heijne J, Sadkowska-Todys M, Rosinska M, Edmunds WJ. Social Contacts and Mixing Patterns Relevant to the Spread of Infectious Diseases. *PLoS Med.* 2008; 5: e74. (https://doi.org/10.1371/ journal.pmed.0050074) PMID: 18366252
- Fumanelli L, Ajelli M, Manfredi P, Vespignani A, Merler S. Inferring the structure of social contacts from demographic data in the analysis of infectious diseases spread. *PLoS Comput Biol.* 2012; 8: e1002673. (https://doi.org/10.1371/journal.pcbi.1002673) PMID: 23028275
- **33.** Iozzi F, Trusiano F, Chinazzi M, Billari FC, Zagheni E, Merler S, Ajelli M, Del Fava E, Manfredi P. Little Italy: an agent-based approach to the estimation of contact patterns-fitting predicted matrices to

serological data. *PLoS Comput Biol.* 2010; 6: e1001021. (https://doi.org/10.1371/journal.pcbi.1001021) PMID: 21152004

- Melegaro A, Del Fava E, Poletti P, Merler S, Nyamukapa C, Williams J, Gregson S, Manfredi P. Social Contact Structures and Time Use Patterns in the Manicaland Province of Zimbabwe. *PLoS ONE*. 2017; 12: e0170459. (https://doi.org/10.1371/journal.pone.0170459) PMID: 28099479
- Machens A, Gesualdo F, Rizzo C, Tozzi AE, Barrat A, Cattuto C. An infectious disease model on empirical networks of human contact: bridging the gap between dynamic network data and contact matrices. BMC Infect Dis. 2013; 13: 185. (https://doi.org/10.1186/1471-2334-13-185) PMID: 23618005
- Bonmarin I, Santa-Olalla P, Lévy-Bruhl D. Modélisation de l'impact de la vaccination sur l'épidémiologie de la varicelle et du zona. *Rev Epidemiol Sante Publique*. 2008; 56: 323–31. (<u>https://doi.org/10.1016/j.</u> respe.2008.07.087) PMID: 18951741
- Melegaro A, Jit M, Gay N, Zagheni E, Edmunds WJ. What types of contacts are important for the spread of infections?: using contact survey data to explore European mixing patterns. *Epidemics*. 2011; 3:143– 51. (https://doi.org/10.1016/j.epidem.2011.04.001) PMID: 22094337
- Poletti P, Melegaro A, Ajelli M, Del Fava E, Guzzetta G, Faustini L, Scalia Tomba G, Lopalco P, Rizzo C, Merler S, Manfredi P. Perspectives on the impact of varicella immunization on herpes zoster. A model-based evaluation from three European countries. *PLoS One*. 2013; 8:e60732. (https://doi.org/10.1371/journal.pone.0060732) PMID: 23613740
- Guzzetta G, Poletti P, Del Fava E, Ajelli M, Scalia Tomba GP, Merler S, Manfredi P. Hope-Simpson's progressive immunity hypothesis as a possible explanation for herpes zoster incidence data. *Am J Epidemiol.* 2013; 177:1134–42. (https://doi.org/10.1093/aje/kws370) PMID: 23548754
- **40.** van Hoek AJ, Melegaro A, Zagheni E, Edmunds WJ, Gay N. Modelling the impact of a combined varicella and zoster vaccination programme on the epidemiology of varicella zoster virus in England. *Vaccine*. 2011; 29: 2411–2420. (https://doi.org/10.1016/j.vaccine.2011.01.037) PMID: 21277405
- Goeyvaerts N, Hens N, Ogunjimi B, Aerts M, Shkedy Z, Van Damme P, Beutels P. Estimating infectious disease parameters from data on social contacts and serological status. J R Stat Soc Ser C Appl Stat. 2010; 59:255–77. (https://doi.org/10.1111/j.1467-9876.2009.00693.x)
- Santermans E, Goeyvaerts N, Melegaro A, Edmunds WJ, Faes C, Aerts M, Beutels P, Hens N. The social contact hypothesis under the assumption of endemic equilibrium: Elucidating the transmission potential of VZV in Europe. *Epidemics*. 2015; 11:14–23. (https://doi.org/10.1016/j.epidem.2014.12. 005) PMID: 25979278
- Silhol R, Alvarez FP, Arena C, Amoros JP, Flahault A, Hanslik T, Boëlle PY. Micro and macro population effects in disease transmission: the case of varicella. *Epidemiol Infect*. 2010; 138: 482–90. (https:// doi.org/10.1017/S0950268809990896) PMID: 19796448
- 44. Oláh LS. Changing families in the European Union: Trends and policy implications. Analytical paper prepared for the United Nations Expert Group Meeting, "Family policy development: achievements and challenges", New York, May 14–15, 2015. Available from: <u>http://www.un.org/esa/socdev/family/docs/</u> egm15/Olahpaper.pdf. Accessed: 26 October 2017.
- Boëlle PY, Hanslik T. Varicella in non-immune persons: incidence, hospitalization and mortality rates. *Epidemiol Infect.* 2002; 129: 599–606. (https://doi.org/10.1017/S0950268802007720) PMID: 12558344